Author’s response to reviews

Title: Comparison among Perfect-C®, Zero-P®, and plates with a cage in single-level cervical degenerative disc disease

Authors:

Sung Hyun Noh (juwuman12@gmail.com)
Ho Yeol Zhang (hyzhang@nhimc.or.kr)

Version: 4 Date: 21 Nov 2017

Author’s response to reviews:

1. It has come to our attention that throughout the manuscript there is significant text overlap (particularly in the methods and discussion sections) with the following publications:

   http://thejns.org/doi/10.3171/2014.6.SPINE12951

   -> I change the similar sentence and I insert article of Njoku et al. as the citation. Thank you for your comment.

2. In the Ethics approval and consent to participate section of the Declarations, please confirm whether informed consent, written or verbal, was obtained from all participants and clearly state this. If verbal, please state the reason and whether the ethics committee approved this procedure. If the need for consent was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

   -> I change the the paragraph of the Ethics approval and consent to participate section. Thank you for your comment